CMSC 735 & MSWE 609

Assignment 2C

This is the third part of the assignment.  In this step, you will meet as a team to discuss the error

and fault lists.  You must submit:

·
       A final list of the errors in the document

·
       A final fault list, so that you can show the mapping from errors to faults.  (You may add new

                         faults to your list if you desire in this phase, but it is not mandator y.)

As your submissions in this phase must represent the work of all gr oup me mbers, the tea m

should merge the individual lists into a final list.  The team must reach a consensus on what

errors to include in the team list, and the faults to which they map.

Please note that the team meeting will not be effective unle ss the individ ual participants have

already found their own faults and abstracted their own error lists.  The team should not meet

until after the team members have already submitted their individual lists to the TA.

You must mail your team error and fault lists to Ryan (
rjones@cs.umd.edu

) by Oct. 9 at noon .

You should download the template report for m 1.3 from the class ftp site and fill in your faults

and errors as described (if you have any pr oblems downloading, let Ryan know and he will send

an ASCII version of the file via email).  Keep track of how long you spend discussing the lists;

you will need this infor mation to answer some of the que stions at the bottom of the report for m

(these que stions are for statistical purposes only and will not be used in deter mining your grade).